Author's response to reviews

Title: Lack of association of two common polymorphisms on 9p21 with risk of coronary heart disease and myocardial infarction; results from a prospective cohort study

Authors:

Abbas Dehghan (a.dehghan@erasusmc.nl)
Mandy van Hoek (m.vanhoek@erasusmc.nl)
Eric J.G Sijbrands (e.sijbrands@erasusmc.nl)
Ben A Oostra (b.oostra@erasusmc.nl)
Albert Hofman (a.hofman@erasusmc.nl)
Cornelia van Duijn (c.vanduijn@erasusmc.nl)
Jacqueline C.M Witteman (j.witteman@erasusmc.nl)

Version: 7 Date: 12 October 2008

Author's response to reviews:

Concerning Submission of second revision of a manuscript

Dear Editor,

Please find enclosed our manuscript entitled ‘Lack of association of two common polymorphisms on 9p21 with risk of coronary heart disease and myocardial infarction; results from a prospective cohort study’. The title and the first paragraph of the introduction are revised according to the comments. We hope that you will find the revision satisfactory and that you will consider it for publication.

Yours sincerely,

Corresponding author: Prof. Dr. J.C.M. Witteman, PhD
Department of Epidemiology,
Erasmus Medical Center,
P.O. Box 2040, 3000 CA Rotterdam,
The Netherlands
Phone: +31-10-70 44190 / 43365
Fax: +31-10-70 44657
Email: j.witteman@erasusmc.nl